Author's response to reviews

Title: A Feasibility Study of Short Message Service Text Messaging as a Surveillance Tool for Alcohol Consumption and Vehicle for Interventions in University Students

Authors:

Simon C Moore (mooresc2@cardiff.ac.uk)
Katherine Crompton (st09003588@outlook.cardiffmet.ac.uk)
Stephanie van Goozen (vangoozens@cf.ac.uk)
Marianne van den Bree (vandenbreemb@cf.ac.uk)
Julia Bunney (bunneyjl@cf.ac.uk)
Emma Lydall (emma.lydall@wales.nhs.uk)

Version: 4 Date: 16 September 2013

Author's response to reviews:

Very many thanks for continuing to provide critical comments, etc.. The paper has been proof read and substantially revised according to reviewers comments.